



Article The Impact of Signals Transmission on Patients' Choice through E-Consultation Websites: An Econometric Analysis of Secondary Datasets

Adnan Muhammad Shah ^{1,*,†}, Rizwan Ali Naqvi ^{2,†} and Ok-Ran Jeong ^{3,*}

- ¹ Department of Information Technology, University of Sialkot, Sialkot 51310, Pakistan
- ² Department of Unmanned Vehicle Engineering, Sejong University, Seoul 05006, Korea; rizwanali@sejong.ac.kr
- ³ School of Computing, Gachon University, Seongnam 13120, Korea
- * Correspondence: adnanshah486@gmail.com (A.M.S.); orjeong@gachon.ac.kr (O.-R.J.)
- + Adnan Muhammad Shah and Rizwan Ali Naqvi are co-first authors.

Abstract: (1) Background: The COVID-19 pandemic has dramatically and rapidly changed the overall picture of healthcare in the way how doctors care for their patients. Due to the significant strain on hospitals and medical facilities, the popularity of web-based medical consultation has drawn the focus of researchers during the deadly coronavirus disease (COVID-19) in the United States. Healthcare organizations are now reacting to COVID-19 by rapidly adopting new tools and innovations such as e-consultation platforms, which refer to the delivery of healthcare services digitally or remotely using digital technology to treat patients. However, patients' utilization of different signal transmission mechanisms to seek medical advice through e-consultation websites has not been discussed during the pandemic. This paper examines the impact of different online signals (online reputation and online effort), offline signals (offline reputation) and disease risk on patients' physician selection choice for e-consultation during the COVID-19 crisis. (2) Methods: Drawing on signaling theory, a theoretical model was developed to explore the antecedents of patients' econsultation choice toward a specific physician. The model was tested using 3-times panel data sets, covering 4231 physicians on Healthgrades and Vitals websites during the pandemic months of January, March and May 2020. (3) Results: The findings suggested that online reputation, online effort and disease risk were positively related to patients' online physician selection. The disease risk has also affected patients' e-consultation choice. A high-risk disease positively moderates the relationship between online reputation and patients' e-consultation choice, which means market signals (online reputation) are more influential than seller signals (offline reputation and online effort). Hence, market signals strengthened the effect in the case of high-risk disease. (4) Conclusions: The findings of this study provide practical suggestions for physicians, platform developers and policymakers in online environments to improve their service quality during the crisis. This article offers a practical guide on using emerging technology to provide virtual care during the pandemic. This study also provides implications for government officials and doctors on the potentials of consolidating virtual care solutions in the near future in order to contribute to the integration of emerging technology into healthcare.

Keywords: e-consultation; COVID-19; signaling theory; patients' choice; panel data analysis

1. Introduction

In late December 2019, unknown pandemic cases emerged in Wuhan, China, and have become a big public health problem not only for China but countries all over the world [1]. The pandemic was declared coronavirus pathogen (COVID-19) by the World Health Organization (WHO) in mid-February [2]. After the spread of the pandemic to dozens of countries, including the United States (U.S.) [3], where 2,537,636 confirmed cases with 126,203 mortalities were reported at the time of writing (i.e., 27 June 2020) by



Citation: Shah, A.M.; Naqvi, R.A.; Jeong, O.-R. The Impact of Signals Transmission on Patients' Choice through E-Consultation Websites: An Econometric Analysis of Secondary Datasets. *Int. J. Environ. Res. Public Health* 2021, *18*, 5192. https:// doi.org/10.3390/ijerph18105192

Received: 17 March 2021 Accepted: 10 May 2021 Published: 13 May 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). the Centre for Diseases Control and Prevention (CDC) [4]. Obviously, the shortage and overloading of healthcare facilities have been the main concern during the outbreak [5]. In addition, delivering timely and efficient healthcare information and services seemed daunting during this pandemic because of insufficient protective gear, lockdown and the danger of spreading the infection to patients and doctors [6]. In order to reduce and control the spread of the pandemic, hospitals can improve their productivity of medical information systems by replacing traditional information systems with virtual platforms [7]. Virtual platforms such as e-consultation websites offer a novel approach to resolving the difficult situation between doctors and patients. These platforms also provide patients with essential details, help them find physicians with a specific style, assist them in their decision-making process, and make them better prepared for their future online consultation [8].

E-consultation, a revolutionary way to meet rising medical demand, helps users resolve barriers of space and time to provide more options for selecting doctors around the world, and it is becoming more widely used these days [9]. Many countries have implemented online health consultation services in order to offer better treatment to patients. One of the most popular platforms to search health-related information is e-consultation websites [10]. A web-based consultation can be useful to supplement conventional healthcare services and traditional doctor-patient relationships [11]. Scholars have indicated that between 25% and 70% of all patients seeking treatment do not want a face-to-face visit with a healthcare provider due to the high demand for online consulting services [12]. One of the major benefits of this modern form of e-consultation service is the minimization of both waiting time and travel costs [13]. These virtual care services offer a wide range of non-dispensing services, allowing doctors to provide high-quality medical care during the COVID-19 pandemic. This most recent type of virtual care can be incorporated into the healthcare system as a way to improve healthcare delivery quality [14]. It encourages social distancing and aids hospitals and clinics in coping with long wait times and the possibility of disease progression. By reducing physical visits and minimal face-to-face interaction among doctors and patients, virtual care systems will support reducing virus transmission and protecting medical professionals from infection [5]. Currently, COVID-19 is causing several medical facilities to cancel or postpone in-person outpatient medical appointments [15]. These e-consultation websites are also important in minimizing medical costs, enhancing clinical quality and efficacy, increasing the equity of medical services and meeting patient satisfaction. In order to make these e-consultation websites more effective for health consumers, it is necessary to adjust the contents of e-consultation websites to match the specific information needs according to the requirements of patients [16]. Due to the above reasons, e-consultation websites are the precious gold mines for patients regarding different aspects of healthcare during the COVID-19 pandemic [17].

Our research context is worth remembering as the healthcare industry is unique, since information asymmetry is fiercer in the traditional health care environment [18]. In a broader domain, information asymmetry represents a state in which few people have more knowledge than others [19]. Information asymmetry is one of the principal causes of disputes in doctor-patient relationships within conventional medical systems [20]. For instance, a competent doctor tends to convey quality information about the clinical service to patients by providing the best medical facilities. From the perspective of signaling theory in healthcare, (1) information asymmetry between a signaler and receiver occurs when physicians know more than patients, which leads to fraudulent activities by physicians to cheat their patients. (2) A potential conflict of interest between the signaler (physicians) and receiver (patients) may occur when physicians cannot serve the best interest of patients and take advantage of information asymmetry. For example, physicians who run their own clinics or treatment centers may prefer to refer patients to their facilities for needless care and gain financial benefits from doing so [20]. Without satisfying either of these two conditions, there is no need for signals because the problem lies in communication between both parties. When both these conditions are satisfied simultaneously, signals are generated by signalers toward receivers to prove their quality of medical service. A highly competent

physician can communicate credible and quality information to their patients [21]. The worth of these informational signals depends on whether highly competent physicians can isolate themselves from less competent physicians. On e-consultation website, patients (receiver) who are intending to consult doctors, seeking different signals from doctors (signalers) about their health condition [22]. Hence, it is important in the virtual platforms that both doctors and patients should transmit different signals to minimize the information asymmetry by fostering the doctor-patient relationship strength.

In our online consultation context, the role of the signaling mechanism becomes important because doctors use intelligent signaling mechanisms not only to communicate socially with patients but also to show their expertise. This strategy is in line with the signaling literature in game theory [19]: by sending signals, one party can credibly pass on some private information about itself to another party. E-consultation websites display both online and offline signals that users can observe to understand the physician quality [23]. Online signals are obtained from virtual consultation-based platforms, including online reputation (market signals) and online effort (seller signals). In contrast, offline signals are derived from the offline setting, such as offline reputation (seller signals). The seller signal is the conventional signal clearly described in the previous signaling studies that the seller deliberately transmits to the buyer [23]. On the other hand, the market signal is the indicator or the information exchanged unconsciously or unknowingly between different players in the market [16]. The market signal as online word of mouth (WOM) is information regarding different products and services created by buyers based on the consumption experience. An example of the market signal is the information flow motivating the herding attitude of stakeholders [24].

Patients and physicians display different signals on e-consultation websites, affecting patients' cognitive abilities to choose a good doctor [16]. People obtain information regarding the quality of healthcare services rendered by physicians through different signaling mechanism [25], such as the process that the signaler (i.e., the physician) sends measurable signals (i.e., offline reputation as physician title, education, experience and online effort) to the recipient (i.e., patients) to communicate details about the non-observable qualities (i.e., competency) in order to minimize information asymmetry [21,26]. Earlier research indicated that receivers are sovereign decision-makers [27]. Therefore, these virtual platforms permit patients to express their satisfaction and dissatisfaction toward a signaler by writing positive or negative reviews [28]. The patients' feedback is reflected as the affective reactions of peers toward their experience with the quality of service that s/he gets from that doctor [29]. For these reasons, it is also important to consider the role of online signals (i.e., online reputation and online effort) and offline signals (offline reputation) generated by markets and sellers on patients' choice of a physician on e-consultation platforms. Based on the assessment of these two measures, patients make more rational decisions while selecting their doctor online for e-consultation.

Compared with face-to-face consultation, e-consultation websites could provide patients with timelier and more convenient information by displaying different signals that could be helpful in patients' physician selection choice for online consultation [30,31]. These signals include physician online reputation, offline reputation and online efforts. For example, Shah, et al. [32] found that online reputation in the form of patient-generated signals positively impacts the patients' online decision-making process. A study by Liu, et al. [33] do demonstrate that individual physician offline and online reputations on e-consultation websites were positively associated with the patients' choice to book appointments online with that particular physician. Hong, et al. [26] indicated that online reputation and knowledge contribution as market signals and seller signals significantly positively influence patients' online choice to pay higher prices to physicians. In another study, the impact of seller signals (self-representation) and market signals (reputation) in e-consultation market is found significantly positive on physician volume of online bookings [16]. Zhang, et al. [34] found that the higher a family doctor's online reputation and knowledge contributions on virtual platforms, the greater that his/her private benefits. Khurana, Qiu and Kumar [20] reported that doctors' online responses as knowledge contribution on online healthcare portal could be regarded as a significant signal of clinical service quality. Furthermore, Wu and Lu [35] indicated that physician online reputation virtual platforms reflect as a signal of quality. Deng, et al. [36] revealed that patients' choice is significantly affected by the physician reputation and effort online. Greenwood, et al. [37] showed that physician offline reputation in the form of high-level expertise significantly influences physician behavior to adopt new information in an online environment. Wu and Deng [38] revealed the positive correlation between physician capital as offline reputation and patients' choice of online physician selection. In addition, physician reputation and benevolence as an effort online were found positively related to the patient assessment of physician performance [39]. Moreover, Li, Zhang, Ma and Liu [10] also indicated physician online reputation and self-representation as information signals in the e-consultation market. Several other studies also indicated that online reputation in the form of online physician reviews affects patients' choice to choose the best doctor online [40–43]. Keeping in view the notion behind signaling theory, signalers (physicians) transmit signals to receivers (patients). Patients can then evaluate these signals to choose an expert healthcare provider who can provide them quality health services.

Based on the earlier discussion, the following research gaps are identified in the previous literature regarding the signaling applied in the online healthcare market. First, as mentioned above, many studies focused on the effects of different signals transmission on patients' choice for online consultation in the pre-COVID-19 scenario. Since virtual treatment has been proposed as a way to maintain quality of care for patients due to a lack of other options; therefore, virtual treatment has become increasingly popular during the recent pandemic [5]. However, only a few doctors and patients are adequately trained about how to use these digital services efficiently [44]. As a result, guidelines and recommendations are needed regarding the usage of different signals (i.e., online signals or offline signals) to educate both physicians and patients on how to transmit different signals on virtual platforms in order to improve the patients' online consultation choice [16,23,36]. Therefore, this study aims to present a unique problem in the COVID-19 pandemic scenario, which incorporates different types of signals in the same research, that is, online signals and offline signals transmitted by marketers and sellers to investigate their impact on patients' physician selection choice for e-consultation using the theoretical foundation of signaling theory.

Signaling theory is useful in understanding the individuals' attitudes and scale down the knowledge asymmetry [27]. The signaling theory provides a theoretical foundation for how one party uses signals to communicate secrete or restricted information to another party regarding the quality of product or service in order to enable the purchase or exchange [45]. Due to information asymmetry in online markets, it is very important to make full use of the signaling theory. In signaling theory, the cost of a signal is the main signaling component, which contributes to signal values [46]. Our research tends to answer the following first question to advance theory and research on the impact of different signals on patients' decisions about physicians for online consultation during the COVID-19 pandemic.

(1) How do online and offline signals transmitted by marketers and sellers on e-consultation websites affect the patients' physician selection choice for online consultation during the pandemic?

Second, the signaling environment plays a critical role in defining which signal to use [47], and the signal strength is mitigated by the signaling environment where the signal operates [48]. Researchers suggest the impact of signaling will be greater in case of high environmental uncertainty. For example, Siering, et al. [49] examined that the influence of signals on review helpfulness vanishes in case of low information environment uncertainty, whereas relationship prevails in a high information environment uncertainty. Similarly, Lester, et al. [50] reported that top management teams' prestige is more valued for firms who are responsible for the initial public offering in the more uncertain environments than

more certain environments. On the other side, several other scholars revealed that the influence of signaling would be weakened in case of high environmental uncertainty. For example, using a sample from U.S-based firms, Certo, et al. [51] explored the moderating role of uncertainty in the relationship between top management teams and global strategic posture. In virtual treatment, the patient decision-making for online consultation also varies across different environments (disease risk) [52]. Hence, this study develops a new theoretical model and tests the argument for patients' physician selection choice for online consultation across varying environments (i.e., disease risk) during the pandemic. This study proposes our second research question is provided as follows:

(2) Does the patients' perceived disease risk have moderating effects on the relationships between both online signals and offline signals, and patients' physician selection choice for e-consultation during the pandemic?

As it is important for patients to search online for the best physician or review their current one, this work offers an overview of how patients make choices online using a real longitudinal panel dataset collected during the recent coronavirus outbreak in the U.S., which was not done in earlier digital healthcare studies. We also hope that the findings of this study would assist policymakers and providers in developing effective and efficient intervention strategies.

The remaining part of this study is organized as follows: Section 2 contains the research hypotheses, Section 3 explains the methods employed, Section 4 features data analysis and results, Section 5 discusses the major findings and contributions, details the implications as well as research the limitations and future research directions. Finally, Section 6 presents the conclusion of this paper.

2. Research Hypotheses

Based on the earlier evidence, this paper links online reputation and online effort as online signals generated by market and seller. In contrast, we use offline reputation as offline signals to describe seller signals only. The study also proposes that environmental uncertainty directly affects patients' physician selection choice for e-consultation (e-consultation choice from here onward). That also has moderating effects on the relationships between online signals, offline signals and patients' e-consultation choice. The proposed model is shown in Figure 1.



Figure 1. Research model.

2.1. Online Reputation and Patients' E-Consultation Choice

Reputation cues presenting others' influence are also important in consumer purchase decision-making. In online markets, consumers need to judge the seller's ability to complete the purchase process successfully. Consumers evaluate the online information posted by

others, which is a challenging task because participants can belong to any geographical location on the Internet [53]. They tend to perceive online reputation information as useful in influencing their judgments regarding products or services. Consumers have access to limited information about a seller or provider searching for a quality product and potential seller. When consumers do not know each other and do not have prior interaction, online reputation reflects the only information about the seller or provider quality that does not originate from the seller or provider themselves [54]. The market signal, that is, online reputation, refers to information regarding different sellers' products and services created by buyers based on their consumption experience. Information is passed passively or unintentionally between the various actors in a market [55]. Researchers claimed that product or service consumers who are satisfied or dissatisfied with a brand would share their experience through market signals [56].

Online reputation phenomenon can also be extended to the delivery of online healthcare services. The physician's *online reputation* refers to patients' perceptions regarding the physician's assessment online after each interaction [33]. To select a competent physician, healthcare consumers take suggestions from their family members and friends to obtain online WOM information. A number of investigations have debated the impact of physician online reputation, such as the star ratings and textual feedback on patients' choice of physician selection online [16,32,35,43,57,58]. The existing literature validates online WOM as a valuable and efficient channel for disseminating information regarding provider reputation to consumers.

On e-consultation websites, patients have more cues to determine the quality of the online services offered by their physicians [59]. When the online market has so many physicians, consumers can only select those with higher star ratings. Contrary, if the market entirely contains physicians with poor reputations, users will not consider these poorly rated physicians for online consultation [60]. Thus, a market with a strong reputation and highly rated physicians is very critical. Suppose the market contains too many physicians with a high online reputation. In that case, market productivity will be enhanced because of a larger consideration set of credible doctors from which users choose physicians (i.e., the demand for highly reputed and quality physicians is increased). In the same line of thought, market productivity will not be enhanced if the market contains limited, highly rated physicians [10]. Extant literature suggests that the consumers' attitudes toward online health services in the form of ratings affect their purchase intentions [40,41]. According to recent studies, a physician's online reputation is a good indicator to reflect the patients' online behavior [10,16]. Thus, we hypothesize the following:

Hypothesis 1 (H1). *Physician with a strong online reputation is positively associated with patients' e-consultation choice.*

2.2. Offline Reputation and Patients' E-Consultation Choice

Physician *offline reputation* is a patient belief about the ability and expertise of a doctor to provide effective and reliable services [61]. Consumers commonly search for reputable physicians before consulting a physician in the e-health context [32]. Previous literature suggests that physicians display their organizational status on their profile page regarding their titles and technical skills to deal with a particular disease [12,16,38,62]. Patients search offline information to seek technical and other physician features as the paramount criteria to evaluate the provider's performance and choose a competent provider [37]. According to Guo, et al. [63], physician status capital (offline reputation) is positively related to physician online selection choice. A study by Li, et al. [64] indicated that professional capital is important in increasing service demands in the online healthcare market. Moreover, the physician capital leads to enhance performance outcome [38]. Professional capabilities are important in driving professional services to patients [62]. Physicians with higher levels of expertise tend to bring novelty to their practices to attract more patients [37]. Several other studies also discussed the relationship between physician offline reputation and patients'

online choice [16,33]. These studies confirmed the healthcare consumers' perceptions that physician offline reputation is the key to their online decision-making. Physicians with a higher clinical tag and/or academic title signal patients that the physician is formally recognized as reputable. We thus hypothesize that:

Hypothesis 2 (H2). *Physician with a strong offline reputation is positively associated with patients' e-consultation choice.*

The term '*effort*' can be defined as the amount of energy consumed on an act per unit of time, while the length of time spent working and the intensities of the work activities are two important aspects of effort [65]. Several researchers have considered the impact of effort on consumer purchase decisions [66,67].

Due to the intangible nature of services, employees' verbal and nonverbal acts significantly impact the consumers' attitudes toward the quality of service [68]. An employee's effort is more critical in service settings, as customer service quality evaluations are often directly associated with service provider performance [69]. If the employee is deemed to be paying extra attention to their job, then s/he may get a higher rating from its customers [70]. Efforts by the employee may impact consumer purchase intention, which is vital to the overall success of a service organization. The positive impact of the effort will lead to the probability that the customer will shop and buy products in the future. Since online healthcare services belong to the service domain, physicians' efforts related to the services they deliver may influence the patients' perceptions regarding service quality and may alter their decisions and opinions about physicians. The study by Deng, Hong, Zhang, Evans and Chen [36] indicated that patients prefer online consultation with physicians who have higher quantities of effort online. Liang, Luo and WU [65] described a positive relationship between a physician's online effort and patients' choice for online consultation. Physicians' with more online contributions have a significant positive impact on their benefits [40]. Furthermore, the findings from Li, Tang, Yen David and Liu [16] revealed that physician online effort in the form of self-representation is positively related to the patient online selection of a particular physician. Apart from these studies, the findings from several other kinds of research also revealed a significant positive effect of physician online efforts on patients' choice to choose a good doctor online [58,71].

While choosing doctors for online consultation, patients can visit the doctors' profile page and get further information, such as personal blogs, articles written and prior correspondence between healthcare providers and patients. Through this point, patients gain a sense of the physician's efforts in the past, which can influence their attitudes toward the physician and, therefore, increases the probability of selecting a good physician. More effort displayed online by the physicians towards their services offering could increase the probability of patients selecting them for online consultation. Thus, the following hypothesis is proposed:

Hypothesis 3 (H3). *Physician with a higher amount of effort online is positively associated with patients' e-consultation choice.*

2.3. Direct and Moderating Impact of Disease Risk

Disease risk measures the severity of the effects of a disease type. According to the signaling theory in the e-health context, the signaling environment significantly affects patient satisfaction levels with service quality [57,72]. In this research, we tend to differentiate the risk between different types of diseases rather than the differences in risk between different patients. This is basically because the former is supposed to be consistent among various individuals. Since mortality is a key indicator to judge disease risk [32], the notion of disease risk has been discussed previously in medical science research [16,32,35,43,57].

A physician treating high-risk disease can deliver patients more opportunities to evaluate his/her technical skills. Online physician reviews are open content that is visible to the public on virtual platforms. They provide patients with some overview regarding the physician skills, including clinical quality and bedside manners. Suppose reviews are posted by patients who involve in high-risk disease and visited their concerned physicians. In that case, these physicians can provide potential patients with high-quality care to enhance patients' online behavior. For example, patients with a severe illness (i.e., cancer) experience more pain and discomfort than patients with a mild illness (i.e., influenza) [57]. The probable explanation for this is mainly because mortality is a significant predictor for assessing the disease risk, and cancer disease has a higher mortality rate for patients than influenza. Therefore, patients involved in high-risk disease need more competent physicians than those involved in low-risk disease. Hence, we propose:

Hypothesis 4 (H4). A physician who treats high-risk disease has more chances to influence patients' e-consultation choice in comparison to those who treat low-risk disease.

According to the signaling theory, differences in the disease risk level affect the patient perception of the provider's online reputation regarding service quality and satisfaction. Scholars have found that the higher the online reputation level of a physician for high-risk disease, the more his/her service quality and service level acknowledged by the sizeable number of patients who consult online [16]. Patients with high-risk disease bear greater financial and health damages; therefore, they need more assurances of high-quality service from a reputed physician than patients with low-risk disease [39]. For instance, Shah, Yan, Shah, Shah and Mamirkulova [32] indicated that the highly-risk disease strengthens the effect of physician online reputation on a patient decision-making process. For high-risk disease, a physician sends high online reputation signals to help patients choose competent physicians, thus effectively assisting patients in distinguishing skilled physicians [40]. Therefore, in such situation a physician higher online reputation can communicate a signal of competency and quality of services for high-risk disease.

Online reputation is more unbiased and reliable than conventional information from peers [73]. Patients with high-risk disease seek a physician's online reputation (e.g., satisfaction ratings and patients' feedback comments) to evaluate the physician's quality of care [43]. Moreover, patients may find more knowledge about physicians' online reputation to measure their service delivery process (e.g., physicians' and office/staff performance), which is critical in affecting patients' online consultation decisions. Several other studies have also reported the moderating effect of disease risk on patients' choice [35,39]. In addition, since the disease risk is a critical factor in patients' decision-making, we assume patients with high-risk diseases will strengthen this motivation between online reputation and patients' e-consultation choice. Therefore, based on the above discussion, we propose:

Hypothesis 5 (H5). *Disease risk moderates the effect of a physician online reputation on patients' e-consultation choice. High-risk disease exhibits a stronger positive connection between the online reputation and e-consultation choice compared to low-risk disease.*

Reputation reflects as a signal of quality. Previous investigations of the signaling theory described that the signaling environment plays a key role in defining the usage of signals [16], and the signaling environment moderated the signal strength [49]. The disease risk in the virtual platforms relates to the purchase significance of healthcare services for patients. In the case of low-risk disease, any physician can treat his/her patients with a high possibility of better care. Therefore, the strength of the relationship between a physician quality (offline reputation) and treatment outcome likely to be low. Conversely, in the case of high-risk disease, the probability that any physician can treat the patient becomes low, and in this situation, the treatment outcome largely depends on the physician's quality (offline reputation). The association between physician quality (offline reputation) and treatment outcome largely depends on the physician's quality (offline reputation). The association between physician quality (offline reputation) and treatment outcome largely depends on the physician's quality (offline reputation). The association between physician quality (offline reputation) and treatment outcome largely depends on the physician's quality (offline reputation). The association between physician quality (offline reputation) and treatment outcome will be strong enough in this situation. Moreover, high-risk diseases are commonly linked with unknown circumstances such as sudden deaths, higher treatment cost burden and lengthy hospital stays. Therefore, patients suffering from high-risk disease

tend to be more sensitive to the physical status as an offline reputation than those suffering from a low-risk disease. Hence, we hypothesize:

Hypothesis 6 (H6). *Disease risk moderates the effect of a physician offline reputation on patients' e-consultation choice. High-risk disease exhibits a stronger positive connection between the offline reputation and e-consultation choice compared to low-risk disease.*

The disease risk may also moderate the relationship between online effort and patients' e-consultation choice. Physicians treating high-risk diseases require a high level of online expertise. Therefore, they can put more effort into online knowledge sharing on e-consultation platforms. With such expertise and skills, doctors will be more willing to exchange free health information online through their personal desires and internal ambitions (social motivation), increasing their financial results by attracting more patients [71]. However, those treating low-risk diseases may put little online effort due to less information-sharing experience on platforms. As a result, relying on their professional motivation to encourage more efforts to share health information is weaker for physicians treating low-risk diseases than physicians treating high-risk diseases [16]. Based on the above arguments, when the disease risk is low, the effect of online effort on patients' econsultation choice will be low because it does not significantly reduce uncertainty. In contrast, when the disease risk is high, the effect of online efforts on patients' econsultation choice should be high since it significantly reduces the associated uncertainty. Hence, we propose that:

Hypothesis 7 (H7). Disease risk moderates the effect of a physician online effort on patients' *e-consultation choice. High-risk disease exhibits a stronger positive connection between the online effort and e-consultation choice compared to low-risk disease.*

3. Methods

3.1. Research Contexts

Our research contexts are two leading commercial online consultation platforms in the U.S, Healthgrades.com and Vitals.com. We construct a novel longitudinal data set using several data sources to test our hypotheses. According to the global internet engagement statistics from Alexa.com, Healthgrades and Vitals had a traffic rank of 7502 and 31,388, respectively. As of June 2020, on average, a visitor spends 1:49 and 1:44 in minutes and seconds on each site every day, respectively. In addition, more than 8 million physicians enrolled in these platforms as of June 2020. We checked both these platforms provide similar structure and services. Each physician on both these sites can create a personal profile page. Both these channels display their basic information, which includes: their demographic statistics, professional title, education, experience and area of expertise, board certifications, contributed articles and service records, etc. In addition to this information, a physician can receive feedback from patients on these platforms in the form of ratings as well as text-based reviews.

3.2. Data

A network spider was developed coded in python 3.6 to scrape data from the homepages of 6344 physicians from both these virtual platforms. Data were collected on 30 January, 30 March and 30 May 2020, during epidemic COVID-19, which covers the period from December 2019–May 2020 from the top 25 metropolitan areas in the U.S. We chose these geographical locations based on the highest Internet usage and the number of physicians with active board licenses in the U.S. [74,75]. This scheme of data collection allows us to create a longitudinal panel dataset with a 60-day interval between each time period. The URLs and home page of physicians' data were matched with 3 different time periods. This process revealed that 1420 physicians' data were not available across all 3 periods (in particular, 722 physicians' data were not available in 2 periods and 698 had 1 period missing), there were more than 1 null value across 3 periods for 496 physicians, and 197 physicians having some abnormal values. After removing the missing and abnormal values, our final dataset includes the 4231 physicians' information. Each individual physician selection was non-random in order to avoid including the same provider more than once.

Due to access, time and cost restraints, we restrict the data collection from virtual platforms to the 14 different diseases. Following the disease mortality rate from the Centre for Disease Control and Prevention [76], these 14 diseases cover acute and chronic diseases, as well as both high mortality rate (heart disease, cancer, unintentional injuries, chronic lower respiratory diseases, cerebrovascular diseases, Alzheimer disease and diabetes mellitus) and low mortality rate diseases (influenza and pneumonia, kidney disease, suicide, chronic liver disease and cirrhosis, septicaemia, hypertension and parkinson disease).

3.3. Variables Measures

3.3.1. Dependent Variable

E-consultation choice: Patients' e-consultation choice is measured by two dimensions. Patients' quantity, who had received online consultations in the last 6-months interval, and the ratio of satisfied patients to all patients. Both these dimensions were averaged to obtain a composite variable.

3.3.2. Independent Variables

Online reputation: Online reputation is measured by two dimensions. Review volume (i.e., the number of reviews for that physician) and rating score (on a scale of 1–5, with 1 being the lowest status and 5 being the highest).

Offline reputation: Offline reputation is measured by four dimensions. Physician professional title in the hospital, education, experience and board certifications. We thus calculate the mean value of four dimensions as a measurement of physician offline reputation.

Online effort: The online effort is measured by the number of blogs that the physician has initiated, the number of physician's health care articles, physician number of replies to patients, and the length of physician reply. Thus, the physician responsiveness component of online efforts could be calculated by the average text length of replies to each review, see Formula (1), where length (rp_{ij}) is the text length of replies to the j_{th} review r_{ij} by physician i, and n is the total number of reviews on the physician i profile page.

$$Online \ Effort_i = \frac{1}{n} \sum_{j=1}^{n} length(rp_{ij}) \tag{1}$$

3.3.3. Moderator

Disease risk (Risk): The disease risk is measured by the severity of the physical and physiological effects. We used the following strategies for classifying the 14 diseases into different risk levels. If the illness is mortally severe, it is considered a high-risk disease. In contrast, low-mortally illness is categorized as a low-risk disease.

Control variables: Control variables in this study include *city rank*–whether physician belongs to first, second or third-tier city according to Internet usage and board-certified physicians, *specialty expertise*–the number of diseases that a physician treats, and *new patients*–whether physician accepts new online appointment–bookings or not. Variables and their description are listed in detail in Table 1. A log transformation was used for quantity, volume, experience, blogs, articles, length and phy_exp because its numerical value was much greater than other values. In some distributions, the log-transform reduces skewness, particularly when large outliers are present.

Variable	Symbol, Description and Measurement	Mean	Std.	Min.	Max.
Dependent variable <i>E-consultation choice</i>					
Patient quantity	Quantity–number of patients consulted a physician online in the last 6-months (logarithmic value).	145.78	543.17	0	15,459
Patient satisfaction	Satisfied-the ratio of satisfied patients to all patients (Original data)	0.11	0.15	0.04	0.21
Independent variable Online reputation					
Review volume	Volume–number of reviews for a particular physician (Logarithmic value).	34.39	17.23	1	123
Rating score	Heat-voting heat for the physician, evaluated by patients (Ordinal)	4.41	0.55	1	4.59
Offline reputation					
Professional title	CTitle–physician clinical title (Dummy: 1–medical doctor/0–otherwise)	0.91	0.49	0	1
Elite education	Education-physician graduated from top-50 medical school (Dummy: 1-yes/0-no)	0.31	0.11	0	1
Tenure	Experience–physician experience (in years) (Logarithmic value)	16.89	5.31	1	51
Board certification	Certification-physician board certification (Dummy: 1-yes/0-no)	0.88	0.21	0	1
Online effort					
Blogs	Blogs–the number of blogs initiated by a physician (Logarithmic value)	3.31	21.2	0	27
Scientific articles	Articles-the number of articles published by a physician (Logarithmic value)	0.12	10.11	0	47
Replies	Replies-the number of replies to patients by physician	5.6	113.2	0	47
Reply length	Length–number of words in a reply (Logarithmic value)	16.45	9.21	39.21	98.72
Moderator					
Disease risk	Risk–measured as disease mortality (Dummy: 1–high-risk disease/0–otherwise)	0.77	0.123	0	1
Control variables					
City rank	Rank–city rank where the physician works. A scale of 1–3 is used, with 1 being the lowest and 3 the highest (Ordinal).	2.55	0.04	1	3
Specialty expertise	Phy_Exp-the number of diseases that the physician is good at curing (Logarithmic value).	52.41	31.12	1	13
New patients	New-whether the physician accepts new patients or not (Dummy: 1-yes/0-no).	0.87	0.12	0	1

Table 1. Variables' measurements.

3.3.4. Estimation Model

Our empirical econometric model fit the e-consultation choice, online reputation, offline reputation, online effort, disease risk, city rank, specialty expertise and new patients' data to the following log-linear relationship. The dependent variables and independent

continuous variables have been converted into a log form, as the distribution may not be normal.

 $log(E - consultation choice_i) = \beta_0 + \beta_1 Rank_i + \beta_2 log(Phy_Exp_i) + \beta_3 New_i + \beta_4 log(Online reputation_i) + \beta_5 Offline reputation_i + \beta_6 Online effort_i + \beta_7 Risk_i + \beta_8 log(Online reputation_i) \times Risk_i + \beta_9 Offline reputation_i \times Risk_i + \beta_{10} log(Online effort_i) \times Risk_i + \varepsilon_i$ where i = 1, 2, ..., n index all physicians $\beta_0 - \beta_{10}$ are the parameters to be estimated ε_i is the error term (2)

3.4. Data Analysis Procedure

All our empirical analyses were run using STATA 12.0 (StataCorp, College Station, TX, USA). We used linear regression for time-series data. To validate the interaction effects results, we followed the procedures from Aiken and West [77]. Furthermore, a different method was used to check the robustness of e-consultation choice.

4. Results

Descriptive statistics for studied variables are listed in Table 1. Pearson correlation for the key variables used in this study is shown in Table 2. Results show that the values of variance inflation factor (VIF) statistics for every independent variable were below the threshold (i.e., 10); hence multicollinearity is not a serious issue in this study, and results are reliable.

Table 2. Variables' correlations.

Variables	1	2	3	4	5	6	7	8
1. E-consultation choice	1.00							
2. Online reputation	0.45	1.00						
3. Offline reputation	0.141	0.062	1.00					
4. Online effort	0.51	0.58	0.55	1.00				
5. Risk	0.191	0.245	0.185	0.188	1.00			
6. Rank	0.32	0.28	0.32	0.02	0.12	1.00		
7. Phy_Exp	0.32	0.41	0.25	0.21	0.23	0.28	1.00	
8. New	0.12	0.14	0.21	0.23	0.20	0.29	0.32	1.00

4.1. Hypotheses Testing Results

The regression results are presented in Table 3. Equations are shown in a hierarchical order. The effects are first shown with only control variables in Model 1, Model 2 to test direct effects and Model 3 tests interaction effects. The adjusted R² and F-value both represent a good fit.

For Model 1, it is observed that both physician experience ($\beta = 0.033$, p < 0.05) and new patients ($\beta = 0.010$, p < 0.01) have positive impacts onto the patients' e-consultation choice, whereas city rank ($\beta = -0.016$, p < 0.01) has a negative influence onto the patients' e-consultation choice.

In model 2, two kinds of signals are explored in this study, such as physician's online reputation as market signals, whereas physician's offline reputation and online effort as seller signals. The two signals, namely, physician's online reputation ($\beta = 1.132$, p < 0.05) and online effort ($\beta = 0.069$, p < 0.001), are positively related to patients' e-consultation choice, thus hypotheses H1 and H3 are supported. The relationship between physician's offline reputation and patients' e-consultation choice is insignificant; the hypothesis H2 could not be confirmed. In addition, the disease risk positively influences the patients' e-consultation choice ($\beta = 0.243$, p < 0.05), thus hypothesis H4 is supported.

Variables	Model 1	Model 2	Model 3
Constant	0.242 ***	0.267 ***	0.287 ***
Constant	(0.112)	(0.102)	(0.091)
Denk	-0.016 **	-0.041 ***	-0.031 ***
Kalik	(0.011)	(0.003)	(0.002)
Dhy, Eve	0.033 *	0.014 *	0.013 *
Fny_Exp	(0.005)	(0.001)	(0.001)
Now	0.010 **	0.016 **	0.025 **
INEW	(0.015)	(0.027)	(0.021)
Online reputation		1.132 *	1.117 *
Olline reputation		(0.006)	(0.009)
Offling reputation		0.067	0.081
Onnie reputation		(0.015)	(0.019)
Online offert		0.069 ***	0.071 ***
Online enort		(0.019)	(0.021)
Diale		0.243 *	0.317 *
NISK		(0.041)	(0.114)
Online reputation × Pick			0.128 *
Online reputation × Kisk			(0.059)
Offling reputation V Pick			0.026
Online reputation × Kisk			(0.011)
Online offert × Rick			0.030
Offinite enort × Kisk			(0.021)
Adjusted-R ²	0.208	0.217	0.230
Log-likelihood ratio	429.631	419.765	411.145
F	76.683 ***	7.174 ***	4.162 ***
Ν	4231	4231	4231

Table 3. Regression results.

Note: Standard errors are in parentheses. * p < 0.05; ** p < 0.01; *** p < 0.001.

Model 3 analyzed the moderating role of disease risk. The interaction term (disease risk) between both physician's offline reputation and patients' e-consultation choice ($\beta = 0.026$, p > 0.05), and physician online effort ($\beta = 0.030$, p > 0.05) and patients' e-consultation choice are not significant; thus, H6 and H7 are not supported. In contrast, the interaction term between physician's online reputation and patients' e-consultation choice ($\beta = 0.128$, p < 0.05) is positive and significant. This means H5 is supported.

The interaction effect in Figure 2 shows that physicians with a higher online reputation (in dashed red line) have a larger slope than those with a lower online reputation (in solid blue line), indicating the positive interaction between physician's online reputation and patients' e-consultation choice. At high-risk disease (in dashed red line), online effort increases more rapidly than at low-risk disease (in solid blue line), indicating that high-risk disease increases the positive effect of physician's online effort on patients' e-consultation choice. This figure provides an extra indication to support H5.

4.2. Robustness Check

To check the robustness of patients' e-consultation choice, we collected a new dataset when the number of new infection cases and mortality rates due to COVID-19 was at its peak (16 June–30 June 2020) [78]. Our empirical analysis was run on this new dataset covering physician homepage information for 15 days' time period. The robust empirical results in Table 4 are consistent with the main results in Table 3.



Figure 2. Interaction effect.

Table 4. Robustness results.

Variables	Model 4	Model 5	Model 6
Constant	0.215 ***	0.232 ***	0.272 ***
Constant	(0.141)	(0.114)	(0.098)
Dank	-0.013 **	-0.036 ***	-0.027 ***
Kalik	(0.015)	(0.008)	(0.005)
Phy Evp	0.029 *	0.012 *	0.011 **
гиу_Ехр	(0.007)	(0.004)	(0.004)
Nour	0.008 **	0.015 ***	0.021 **
INEW	(0.019)	(0.034)	(0.026)
Online reputation		1.115 **	1.109 *
Online reputation		(0.014)	(0.015)
Offling reputation		0.041	0.071
Omme reputation		(0.025)	(0.027)
Online offert		0.057 ***	0.065 ***
Olline enon		(0.024)	(0.024)
Diale		0.219 *	0.219 **
RISK		(0.059)	(0.137)
Online reputation V Pick			0.114 *
Online reputation × Kisk			(0.071)
Offling reputation × Pick			0.021
Online reputation × Kisk			(0.017)
Online offert v Pick			0.026
Offinite enort × Kisk			(0.029)
Adjusted-R2	0.241	0.262	0.291
Ν	821	821	821

Note: Standard errors are in parentheses. * p < 0.05; ** p < 0.01; *** p < 0.001.

5. Discussion

The COVID-19 pandemic has triggered public health concerns around the globe. In certain special cases (e.g., the COVID-19 crisis), e-consultation could accomplish the target of public isolation and fulfill the patients' needs for high-quality care [79]. In comparison to physical consultation, e-consultation can only be used to diagnose mild and high-risk

diseases in the absence of appropriate medical exams and treatment systems. However, written recommendations from physicians are still useful to the general public [80]. While e-consultation cannot wholly replace physical consultations, it can be used as a first step after being incorporated into a physical consultation [30]. As a result, while enhancing the functionality of various signaling mechanisms, providers should also improve the patients' choice of physician selection on e-consultation websites [16]. Our findings show that promoting the role of e-consultation is critical to attracting the public to use it, as positive online signals suggest that the public has more confidence in doctors' capacity and honesty.

The main objective of the current study was to develop a theoretical model and examine how different online and offline signals generated by market and sellers predict patients' online consultation choice. The model was tested using 3 points longitudinal panel data obtained from Healthgrades and Vitals websites during the pandemic period in the U.S. As a result, this study presents a number of key findings, theoretical contributions and implications for practitioners.

5.1. Key Findings

The current study presents four major contributions.

First, the positive online reputation significantly and positively influences the patients' physician selection choice for e-consultation. According to the signaling theory, when there are positive online WOM signals about a physician's reputation, new patients are more likely to consult him/her online [16,32]. During the pandemic, patients can enjoy the quality of life and higher-quality online services from a physician with a high online reputation.

Second, the current study result shows the insignificant relationship between offline reputation and patients' e-consultation choice. This finding is different from previous studies, which generally showed the significant and positive effects of the offline reputation [16,26,37]. A possible justification of this result could be travel bans, lockdowns and the possibility of infection spreading during the COVID-19 pandemic; patients rely more on online WOM from peers to seek physician details rather than search for offline information. However, our study findings are supported by a couple of previous studies in which public voice and WOM positively affect social presence [81,82]. Our findings showed patients would prefer those physicians for online consultation, with a high online social presence relatively high reputation in offline hospitals. Results suggest that physicians can work hard online to improve their reputation and build a strong tie with their patients.

Third, the online effort is positively related to patients' e-consultation choice. This finding indicates that physicians willing to spend more time and effort online to benefit patients will eventually get more patients [18,36]. Even though many physicians are registered on virtual platforms to provide online consultation services to their patients during the pandemic, the public is always keen to try those physicians who put more effort online. Therefore, it is essential to strengthening the publicity of e-consultation for the public. In the promotion of physicians on e-consultation platforms, e-consultation providers should offer advanced services for innovative and stable services for the common [79].

Fourth, disease risk is positively related to patients' e-consultation choice. As mentioned earlier, regarding the current situation of the epidemic, travel restrictions and quarantine rules, when the risk of disease is high, patients are more likely to seek health information from online platforms rather than offline hospitals. As a result, virtual health platforms will still be the first choice for patients to consult physicians for high-risk disease treatment.

Fifth, the impact of online reputation on patients' e-consultation choice is higher for high-risk disease than for low-risk disease. When the risk of disease is low, the treatment outcome is certainly low in uncertainty, and thus the signalling is less effective. In contrast, when the risk of disease is high, the treatment outcome is high in terms of uncertainty. The key explanation for this argument is that the patient is highly dependent on the physician's quality of online medical services (online reputation). This makes the signalling more efficient for e-consultation [16].

Sixth, according to our findings, the impacts of offline reputation and online effort are not influenced by the disease risk, which is consistent with the findings of previous studies. For example, Li, Tang, Yen David and Liu [16] reported that offline reputation and online effort both differ from online reputation with regard to signalling source. The offline reputation and online effort are the main signals on the e-consultation platform transmitted by the sellers (i.e., seller signals), whereas the service popularity is the signal initiated by the buyers (i.e., market signals). The study results reveal that the disease risk moderates the impacts of market signals but does not affect seller signals. This is due to the fact that market signals are the direct input from previous patients on the quality of online services. In contrast, seller signals are usually the information that is effectively transmitted to patients by the physicians. Patients have to judge the quality of online services regarding offline reputation and online effort sent by the physicians themselves. Notably, in terms of service efficiency, market signals are more efficient than seller signals. Hence, the influence of market signals on e-consultation platforms becomes more critical when the risk of disease becomes high, and the effects of seller signals do not rely on the level of disease risk.

5.2. Theoretical Implications

This research also provides several theoretical contributions to the existing research. First, we utilize the information asymmetry theory to frame our hypotheses in the online health services context. Accordingly, we systematically analyzed and incorporated all the individual signals in single research (i.e., online and offline signals), which affect patients' consultation choice in the online healthcare market. In particular, we considered online reputation and online effort as online signals, whereas offline reputation as offline signals. Even though previous studies have explored the impact of two or more signals in online healthcare market [18,33,36,37,63]; however, all these market and seller generated signals were not scientifically investigated in a single study, and hence, we cannot compare their effects explicitly because of the different laboratory conditions and independent study outcomes. Thus, our results may improve the treatment outcome for individuals by providing accurate information about the healthcare provider during COVID-19.

Second, this study extends the research on the concept of environmental uncertainty in understanding patients' physician selection choice for e-consultation. Referring to the literature in strategic management, the concept of environmental uncertainty is claimed as the disease risk in this research. Therefore, this research explored the moderating impact of disease risk on different signaling mechanisms on e-consultation websites (online reputation, offline reputation and online effort, etc.), which, according to our information, were not tested in earlier studies. The outcome of this analysis reveals the moderating impact of disease risk onto the online reputation signals on the e-consultation platforms, suggesting that there is a higher effect of online reputation for high-risk disease. This result means the signals are not uniformly useful across all disease types. The online reputation signals would be ideal and best-suited for high-risk disease.

Third, by incorporating variables related to healthcare into the research model, this research also provides evidence for information systems and healthcare researchers. Even though online reputation, offline reputation and online effort signals were debated in previous online healthcare signaling literature; however, their differences were rarely investigated. This study empirically examined that online reputation signals should be treated differently from both offline reputation signals and online effort signals. In fact, online reputation is the market signal transmitted by the buyers in the online market. In contrast, offline reputation and online effort relate to the seller signals transmitted by the healthcare providers themselves. This work, therefore, adds to previous literature on signaling and related fields by demonstrating that market signals may be more predictive than seller signals. Hence these signals tend to be more influential when there is a high risk of disease. As a result, it is important to adjust the policies to adapt to various disease categories.

Fourth, methodologically this study contributes to the literature [12,36] by using a large longitudinal panel dataset in three different time periods that combines intelligent signaling mechanism, such as online and offline signals; to examine how the combination of different signals effect patients' e-consultation choice during the COVID-19 pandemic. To understand patients' behavior, the current study collected real information from physician home pages during the pandemic, which, according to our knowledge, was not investigated in earlier studies.

5.3. Practical Implications

Our study has some substantial practical implications.

For physicians, this study indicates that they may benefit from using e-consultation platforms; in the current pandemic crisis, they should be motivated to use these platforms continuously. Our results reveal differences between e-consultation platforms and conventional e-commerce websites. From a practical point of view, an e-commerce website is a forum for supportive product or service information, with feedback mechanisms provided by certain websites. However, maintaining the strong ties between providers (doctors) and buyers (patients) are more important than conveying information online. Particularly, regarding the knowledge sharing as an online effort by physicians on e-consultation platforms, this study highlights that by using virtual platforms, physicians can improve doctor-patient contact to enhance patients' attitudes toward a provider during the COVID-19 pandemic situation. This evidence would be further strengthened for junior physicians with low status and investing more time and achieving a reputation for their career. This is also a significant factor in improving doctor communication skills, such as developing incentive structures to strengthen the doctor relationship with his/her current patients, which is more beneficial for physicians than recruiting new patients.

For platform developers and policymakers, this study shows that online reputation and online effort factors can affect patients' physician selection choice for e-consultation. Thus, website managers can develop convincing windows on different e-consultation platforms and include all these necessary components to develop an effective e-consultation platform. As a result, efficient signals can be conveyed by these components. From a policy perspective, the current study suggests that e-consultation platforms be required to provide reliable data on doctor-patient interactions during COVID-19. Our findings also show that the signals are not globally reliable, with a stronger impact on high-risk disease.

To end with, the disease risk has a positive moderating role in the relationship between online reputation and patients' e-consultation choice, an understanding of what quality and value mean to patients with different disease risk offers the promise of improving the physician's brand position through precise market analysis and segmentation, service planning and pricing strategy. A physician that treats high-risk disease patients should devote more attention to online reputation building in order to improve patients' attitudes towards him/her. For example, a physician could improve his/her online service attitude, improve response speed and provide detailed information on different aspects of healthcare.

5.4. Limitations and Future Research

Future efforts can be focused on several possible extensions of this work. First, we have only chosen a certain duration of the outbreak to evaluate patients' online behavior. The pandemic is still at its peak in the U.S., with thousands of new infections being reported during and after the period of data collection, and patterns are changing on a daily basis. With several reports of successful vaccine trials in the U.S. and other countries, we might have missed a few insights from the results. As a scheme of our future research, the data collection will continue to test these variables when the number of new infection cases declines substantially in the U.S. Second, although both websites confirm that the physicians' information on their homepages was actually posted by the physicians themselves the authenticity of this information cannot always be guarantee; future research could cross-validate the

physician information using other websites such as https://www.certificationmatters.org, https://www.cms.gov/Physician-Compare-Initiative/ and https://www.jmr.fsmb.org/.

6. Conclusions

Analyzing patients' online behavior using data collected during the novel coronavirus shed light on how the patients use multiple signals to consult a particular physician online. This study is the first step to understanding the impact of online and offline signals on patients' physician selection choice for e-consultation during the COVID-19 crisis. Based on the signaling theory, physician online reputation and online effort were identified as antecedents of patients' e-consultation choice. In addition, disease risk has a positive effect on patients' e-consultation choice. Moreover, a positive effect was identified on the online reputation, demonstrating a higher effect for high-risk disease. These findings validate the claim during the recent pandemic crisis, patients are more motivated to search physician information from online channels rather than offline channels. Finally, several suggestions are provided for physicians, platform developers and policymakers to manage the necessary signaling components on e-consultation platforms.

Author Contributions: Conceptualization, R.A.N.; methodology, A.M.S.; software, A.M.S.; validation, A.M.S. and R.A.N.; formal analysis, A.M.S.; investigation, R.A.N.; resources, A.M.S.; data curation, R.A.N.; writing—original draft preparation, A.M.S.; writing—review and editing, A.M.S.; visualization, R.A.N.; supervision, O.-R.J.; project administration, O.-R.J.; funding acquisition, O.-R.J. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the Basic Science Research Program through the NRF (National Research Foundation of Korea), funded by the MSIT (Ministry of Science and ICT), Korea (Grant No. NRF2019R1A2C1008412).

Institutional Review Board Statement: Not Applicable.

Informed Consent Statement: Not Applicable.

Data Availability Statement: Not Applicable.

Acknowledgments: The authors extend their appreciation to the Basic Science Research Program through the NRF (National Research Foundation of Korea), funded by the MSIT (Ministry of Science and ICT), Korea .

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Meng, L.; Hua, F.; Bian, Z. Coronavirus Disease 2019 (COVID-19): Emerging and Future Challenges for Dental and Oral Medicine. J. Dent. Res. 2020, 99, 481–487. [CrossRef]
- WHO. Coronavirus Disease (COVID-19) Outbreak. Available online: https://www.who.int/emergencies/diseases/novelcoronavirus-2019 (accessed on 30 June 2020).
- 3. Schuchat, A.; Covid, C.; Team, R. Public Health Response to the Initiation and Spread of Pandemic COVID-19 in the United States, February 24–April 21, 2020. *Morb. Mortal. Wkly. Rep.* **2020**, *69*, 551–556. [CrossRef]
- CDC. Coronavirus Disease 2019 (COVID-19). Available online: https://www.cdc.gov/coronavirus/2019-ncov/cases-updates/ cases-in-us.html (accessed on 27 June 2020).
- 5. Jnr, B.A. Use of Telemedicine and Virtual Care for Remote Treatment in Response to COVID-19 Pandemic. J. Med. Syst. 2020, 44, 132. [CrossRef]
- 6. Golinelli, D.; Boetto, E.; Carullo, G.; Nuzzolese, A.G.; Landini, M.P.; Fantini, M.P. How the COVID-19 pandemic is favoring the adoption of digital technologies in healthcare: A literature review. *medRxiv* **2020**. [CrossRef]
- Neubeck, L.; Hansen, T.; Jaarsma, T.; Klompstra, L.; Gallagher, R. Delivering healthcare remotely to cardiovascular patients during COVID-19:A rapid review of the evidence. *Eur. J. Cardiovasc. Nurs.* 2020, 19, 486–494. [CrossRef]
- Li, L.; Liu, G.; Xu, W.; Zhang, Y.; He, M. Effects of Internet Hospital Consultations on Psychological Burdens and Disease Knowledge During the Early Outbreak of COVID-19 in China: Cross-Sectional Survey Study. J. Med. Internet Res. 2020, 22, e19551. [CrossRef]
- 9. Gong, K.; Xu, Z.; Cai, Z.; Chen, Y.; Wang, Z. Internet Hospitals Help Prevent and Control the Epidemic of COVID-19 in China: Multicenter User Profiling Study. J. Med. Internet Res. 2020, 22, e18908. [CrossRef]

- 10. Li, J.; Zhang, Y.; Ma, L.; Liu, X. The Impact of the Internet on Health Consultation Market Concentration: An Econometric Analysis of Secondary Data. *J. Med. Internet Res.* **2016**, *18*, e276. [CrossRef] [PubMed]
- 11. Zhang, X.; Guo, X.; Lai, K.-H.; Yi, W. How does online interactional unfairness matter for patient–doctor relationship quality in online health consultation? The contingencies of professional seniority and disease severity. *Eur. J. Inf. Syst.* **2019**, *28*, 336–354. [CrossRef]
- 12. Li, Y.; Yan, X.; Song, X. Provision of Paid Web-Based Medical Consultation in China: Cross-Sectional Analysis of Data From a Medical Consultation Website. *J. Med. Internet Res.* **2019**, *21*, e12126. [CrossRef] [PubMed]
- Chen, L.; Rai, A.; Guo, X. Physicians' Online Popularity and Price Premiums for Online Health Consultations: A Combined Signaling Theory and Online Feedback Mechanisms Explanation. In Proceedings of the Thirty Sixth International Conference on Information Systems, Fort Worth, TX, USA, 13–16 December 2015; pp. 2105–2115.
- 14. Hassounah, M.; Raheel, H.; Alhefzi, M. Digital Response During the COVID-19 Pandemic in Saudi Arabia. *J. Med. Internet Res.* **2020**, 22, e19338. [CrossRef]
- 15. Bryant, M.S.; Fedson, S.E.; Sharafkhaneh, A. Using Telehealth Cardiopulmonary Rehabilitation during the COVID-19 Pandemic. *J. Med. Syst.* **2020**, *44*, 125. [CrossRef] [PubMed]
- 16. Li, J.; Tang, J.; Yen David, C.; Liu, X. Disease risk and its moderating effect on the e-consultation market offline and online signals. *Inf. Technol. People* **2019**, *32*, 1065–1084. [CrossRef]
- 17. Murphy, M.; Scott, L.J.; Salisbury, C.; Turner, A.; Scott, A.; Denholm, R.; Lewis, R.; Iyer, G.; Macleod, J.; Horwood, J. Implementation of remote consulting in UK primary care following the COVID-19 pandemic: A mixed-methods longitudinal study. *Br. J. Gen. Pract.* 2021, 71, e166–e177. [CrossRef]
- 18. Laugesen, J.; Hassanein, K.; Yuan, Y. The Impact of Internet Health Information on Patient Compliance: A Research Model and an Empirical Study. *J. Med. Internet Res.* **2015**, *17*, e143. [CrossRef] [PubMed]
- 19. Spence, A.M. *Market Signaling: Informational Transfer in Hiring and Related Screening Processes (Harvard Economic Studies)*, 1st ed.; Harvard University Press: Cambridge, MA, USA, 1974; Volume 143.
- 20. Khurana, S.; Qiu, L.; Kumar, S. When a Doctor Knows, It Shows: An Empirical Analysis of Doctors' Responses in a Q&A Forum of an Online Healthcare Portal. *Inf. Syst. Res.* **2019**, *30*, 872–891. [CrossRef]
- 21. Shah, A.M. Research on the Impact of Online and Offline Signals Transmission in Leveraging Physician Economics Returns. Ph.D. Thesis, Published at School of Management, Harbin Institute of Technology, Harbin, China, 2021.
- 22. Chen, L.; Baird, A.; Straub, D. A linguistic signaling model of social support exchange in online health communities. *Decis. Support Syst.* **2020**, *130*, 113233. [CrossRef]
- 23. Li, J.; Tang, J.; Jiang, L.; Yen, D.C.; Liu, X. Economic Success of Physicians in the Online Consultation Market: A Signaling Theory Perspective. *Int. J. Electron. Commer.* 2019, 23, 244–271. [CrossRef]
- 24. Al-Hasan, A. Online Social Stock Picking: An Empirical Examination. Int. J. Electron. Commer. 2018, 22, 66–97. [CrossRef]
- 25. Chen, Q.; Yan, X.; Zhang, T. Converting Visitors of Physicians' Personal Websites to Customers in Online Health Communities: Longitudinal Study. J. Med. Internet Res. 2020, 22, e20623. [CrossRef]
- 26. Hong, Y.A.; Liang, C.; Radcliff, T.A.; Wigfall, L.T.; Street, R.L. What Do Patients Say About Doctors Online? A Systematic Review of Studies on Patient Online Reviews. *J. Med. Internet Res.* **2019**, *21*, e12521. [CrossRef] [PubMed]
- Connelly, B.L.; Certo, S.T.; Ireland, R.D.; Reutzel, C.R. Signaling Theory: A Review and Assessment. J. Manag. 2011, 37, 39–67. [CrossRef]
- 28. Shah, A.M.; Yan, X.; Tariq, S.; Ali, M. What patients like or dislike in physicians: Analyzing drivers of patient satisfaction and dissatisfaction using a digital topic modeling approach. *Inf. Process. Manag.* **2021**, *58*, 102516. [CrossRef]
- 29. Rothenfluh, F.; Schulz, P.J. Content, Quality, and Assessment Tools of Physician-Rating Websites in 12 Countries: Quantitative Analysis. J. Med. Internet Res. 2018, 20, e212. [CrossRef]
- 30. Pecina, J.L.; North, F. Early e-consultation face-to-face conversions. J. Telemed. Telecare 2016, 22, 269–276. [CrossRef]
- 31. Lu, N.; Wu, H. Exploring the impact of word-of-mouth about Physicians' service quality on patient choice based on online health communities. *BMC Med. Inform. Decis. Mak.* **2016**, *16*, 151. [CrossRef] [PubMed]
- 32. Shah, A.M.; Yan, X.; Shah, S.A.A.; Shah, S.J.; Mamirkulova, G. Exploring the impact of online information signals in leveraging the economic returns of physicians. *J. Biomed. Inform.* **2019**, *98*, 103272. [CrossRef] [PubMed]
- 33. Liu, X.; Guo, X.; Wu, H.; Wu, T. The Impact of Individual and Organizational Reputation on Physicians' Appointments Online. *Int. J. Electron. Commer.* **2016**, *20*, 551–577. [CrossRef]
- 34. Zhang, M.; Guo, X.; Wu, T. Impact of Free Contributions on Private Benefits in Online Healthcare Communities. *Int. J. Electron. Commer.* **2019**, *23*, 492–523. [CrossRef]
- 35. Wu, H.; Lu, N. How your colleagues' reputation impact your patients' odds of posting experiences: Evidence from an online health community. *Electron. Commer. Res. Appl.* **2016**, *16*, 7–17. [CrossRef]
- Deng, Z.; Hong, Z.; Zhang, W.; Evans, R.; Chen, Y. The Effect of Online Effort and Reputation of Physicians on Patients' Choice: 3-Wave Data Analysis of China's Good Doctor Website. *J. Med. Internet Res.* 2019, 21, e10170. [CrossRef] [PubMed]
- 37. Greenwood, B.N.; Agarwal, R.; Agarwal, R.; Gopal, A. The Role of Individual and Organizational Expertise in the Adoption of New Practices. *Organ. Sci.* 2019, *30*, 191–213. [CrossRef]
- 38. Wu, H.; Deng, Z. Knowledge collaboration among physicians in online health communities: A transactive memory perspective. *Int. J. Inf. Manag.* **2019**, *49*, 13–33. [CrossRef]

- 39. Yang, M.; Diao, Y.; Kiang, M.Y. Physicians' Sales of Expert Service Online: The Role of Physician Trust and Price. In Proceedings of the Twenty-Second Pacific Asia Conference on Information Systems, Yokohama, Japan, 26–30 June 2018; pp. 1–14.
- 40. Han, X.; Qu, J.; Zhang, T. Exploring the impact of review valence, disease risk, and trust on patient choice based on online physician reviews. *Telemat. Inform.* **2019**, *45*, 101276. [CrossRef]
- 41. Lu, W.; Wu, H. How Online Reviews and Services Affect Physician Outpatient Visits: Content Analysis of Evidence from Two Online Health Care Communities. *JMIR Med. Inform.* **2019**, *7*, e16185. [CrossRef]
- 42. Grabner-Kräuter, S.; Waiguny, M.K. Insights Into the Impact of Online Physician Reviews on Patients' Decision Making: Randomized Experiment. J. Med. Internet Res. 2015, 17, e93. [CrossRef] [PubMed]
- 43. Cao, X.; Liu, Y.; Zhu, Z.; Hu, J.; Chen, X. Online selection of a physician by patients: Empirical study from elaboration likelihood perspective. *Comput. Hum. Behav.* 2017, 73, 403–412. [CrossRef]
- 44. Shankar, M.; Fischer, M.; Brown-Johnson, C.G.; Safaeinili, N.; Haverfield, M.C.; Shaw, J.G.; Verghese, A.; Zulman, D.M. Humanism in telemedicine: Connecting through virtual visits during the COVID-19 pandemic. *Deep Blue* **2020**, preprint.
- 45. Wells, J.D.; Valacich, J.S.; Hess, T.J. What Signal Are You Sending? How Website Quality Influences Perceptions of Product Quality and Purchase Intentions. *MIS Q.* 2011, 35, 373–396. [CrossRef]
- 46. BliegeBird, R.; Smith, E. Signaling Theory, Strategic Interaction, and Symbolic Capital. *Curr. Anthropol.* 2005, 46, 221–248. [CrossRef]
- 47. Ndofor, H.A.; Levitas, E. Signaling the Strategic Value of Knowledge. J. Manag. 2004, 30, 685–702. [CrossRef]
- 48. Park, N.K.; Mezias, J.M. Before and after the technology sector crash: The effect of environmental munificence on stock market response to alliances of e-commerce firms. *Strateg. Manag. J.* **2005**, *26*, 987–1007. [CrossRef]
- 49. Siering, M.; Muntermann, J.; Rajagopalan, B. Explaining and predicting online review helpfulness: The role of content and reviewer-related signals. *Decis. Support Syst.* **2018**, *108*, 1–12. [CrossRef]
- 50. Lester, R.H.; Certo, S.T.; Dalton, C.M.; Dalton, D.R.; Cannella, A.A.J. Initial Public Offering Investor Valuations: An Examination of Top Management Team Prestige and Environmental Uncertainty. J. Small Bus. Manag. 2006, 44, 1–26. [CrossRef]
- Certo, S.T.; Lester, R.H.; Dalton, C.M.; Dalton, D.R. Top Management Teams, Strategy and Financial Performance: A Meta-Analytic Examination. J. Manag. Stud. 2006, 43, 813–839. [CrossRef]
- 52. Chen, S.; Guo, X.; Wu, T.; Ju, X. Exploring the Online Doctor-Patient Interaction on Patient Satisfaction Based on Text Mining and Empirical Analysis. *Inf. Process. Manag.* 2020, *57*, 102253. [CrossRef]
- 53. Djawadi, B.M.; Fahr, R.; Haake, C.-J.; Recker, S. Maintaining vs. milking good reputation when customer feedback is inaccurate. *PLoS ONE* **2018**, *13*, e0207172. [CrossRef]
- 54. Luca, M.; Zervas, G. Fake It Till You Make It: Reputation, Competition, and Yelp Review Fraud. *Manag. Sci.* 2016, 62, 3412–3427. [CrossRef]
- 55. Seiler, S.; Yao, S.; Wang, W. Does Online Word of Mouth Increase Demand? (And How?) Evidence from a Natural Experiment. *Mark. Sci.* 2017, *36*, 838–861. [CrossRef]
- 56. Gopinath, S.; Thomas, J.S.; Krishnamurthi, L. Investigating the Relationship between the Content of Online Word of Mouth, Advertising, and Brand Performance. *Mark. Sci.* 2014, *33*, 241–258. [CrossRef]
- 57. Yang, H.; Guo, X.; Wu, T. Exploring the influence of the online physician service delivery process on patient satisfaction. *Decis. Support Syst.* **2015**, *78*, 113–121. [CrossRef]
- 58. Yang, H.; Zhang, X. Investigating the Effect of Paid and Free Feedback About Physicians' Telemedicine Services on Patients' and Physicians' Behaviors: Panel Data Analysis. *J. Med. Internet Res.* **2019**, *21*, e12156. [CrossRef] [PubMed]
- Fiksdal, A.S.; Kumbamu, A.; Jadhav, A.S.; Cocos, C.; Nelsen, L.A.; Pathak, J.; McCormick, J.B. Evaluating the Process of Online Health Information Searching: A Qualitative Approach to Exploring Consumer Perspectives. *J. Med. Internet Res.* 2014, 16, e224. [CrossRef]
- 60. De Barra, M.; Eriksson, K.; Strimling, P. How Feedback Biases Give Ineffective Medical Treatments a Good Reputation. J. Med. Internet Res. 2014, 16, e193. [CrossRef]
- 61. Shin, D.-H.; Lee, S.; Hwang, Y. How do credibility and utility play in the user experience of health informatics services? *Comput. Hum. Behav.* **2017**, *67*, 292–302. [CrossRef]
- 62. Zhou, J.; Zuo, M.; Ye, C. Understanding the factors influencing health professionals' online voluntary behaviors: Evidence from YiXinLi, a Chinese online health community for mental health. *Int. J. Med. Inform.* **2019**, *130*, 103939. [CrossRef]
- 63. Guo, S.; Guo, X.; Fang, Y.; Vogel, D. How Doctors Gain Social and Economic Returns in Online Health-Care Communities: A Professional Capital Perspective. J. Manag. Inf. Syst. 2017, 34, 487–519. [CrossRef]
- 64. Li, J.; Wu, H.; Deng, Z.; Lu, N.; Evans, R.; Xia, C. How professional capital and team heterogeneity affect the demands of online team-based medical service. *BMC Med. Inform. Decis. Mak.* **2019**, *19*, 119. [CrossRef]
- 65. Liang, Q.; Luo, F.J.; WU, Y.Z. The impact of doctor's efforts and reputation on the number of new patients in online health community. *Chin. J. Health Policy* **2017**, *10*, 63–71.
- 66. Krishnan, B.C.; Netemeyer, R.G.; Boles, J.S. Self-Efficacy, Competitiveness, and Effort as Antecedents of Salesperson Performance. *J. Pers. Sell. Sales Manag.* 2002, 22, 285–295. [CrossRef]
- 67. VandeWalle, D.; Brown, S.P.; Cron, W.L.; Slocum, J.W.J. The influence of goal orientation and self-regulation tactics on sales performance: A longitudinal field test. *J. Appl. Psychol.* **1999**, *84*, 249. [CrossRef]

- 68. Testa, M.R. Organizational Commitment, Job Satisfaction, and Effort in the Service Environment. J. Psychol. 2001, 135, 226–236. [CrossRef]
- 69. Bitner, M.J.; Booms, B.H.; Mohr, L.A. Critical Service Encounters: The Employee's Viewpoint. J. Mark. 1994, 58, 95–106. [CrossRef]
- 70. McClean, E.; Collins, C.J. High-commitment HR practices, employee effort, and firm performance: Investigating the effects of HR practices across employee groups within professional services firms. *Hum. Resour. Manag.* **2011**, *50*, 341–363. [CrossRef]
- 71. Zhang, X.; Guo, F.; Xu, T.; Li, Y. What motivates physicians to share free health information on online health platforms? *Inf. Process. Manag.* **2020**, *57*, 102166. [CrossRef]
- 72. Li, J.; Liu, M.; Li, X.; Liu, X.; Liu, J. Developing Embedded Taxonomy and Mining Patients' Interests From Web-Based Physician Reviews: Mixed-Methods Approach. *J. Med. Internet Res.* **2018**, *20*, e254. [CrossRef] [PubMed]
- 73. Wang, Z.; Li, H.; Ye, Q.; Law, R. Saliency effects of online reviews embedded in the description on sales: Moderating role of reputation. *Decis. Support Syst.* **2016**, *87*, 50–58. [CrossRef]
- 74. Federation of State Medical Boards. Journal of Medical Regulation. Available online: http://jmr.fsmb.org/ (accessed on 21 March 2020).
- FORBES.COM. America's Most Wired Cities: Raleigh Takes the Lead in Our Annual List of the Most Broadband-Connected, U.S. Cities. Available online: https://www.forbes.com/2017/12/31/ (accessed on 26 March 2020).
- 76. 2018. Available online: https://www.cdc.gov/nchs/data/databriefs/db355-h.pdf (accessed on 1 March 2020).
- 77. Aiken, L.S.; West, S.G. *Multiple Regression: Testing and Interpreting Interactions*; Sage Publications, Inc.: Thousand Oaks, CA, USA, 1991; p. xi, 212.
- 78. CDC. COVID Data Tracker. Available online: https://www.cdc.gov/coronavirus/2019-ncov/cases-updates/cases-in-us.html (accessed on 30 June 2020).
- 79. Qi, M.; Cui, J.; Li, X.; Han, Y. Perceived Factors Influencing the Public Intention to Use E-Consultation: Analysis of Web-Based Survey Data. *J. Med. Internet Res.* **2021**, *23*, e21834. [CrossRef]
- 80. Wu, H.; Lu, N. Online written consultation, telephone consultation and offline appointment: An examination of the channel effect in online health communities. *Int. J. Med. Inform.* **2017**, *107*, 107–119. [CrossRef]
- 81. Oh, J.; Ki, E.-J. Factors affecting social presence and word-of-mouth in corporate social responsibility communication: Tone of voice, message framing, and online medium type. *Public Relat. Rev.* **2019**, *45*, 319–331. [CrossRef]
- Yasir, A.; Hu, X.; Ahmad, M.; Rauf, A.; Shi, J.; Ali Nasir, S. Modeling Impact of Word of Mouth and E-Government on Online Social Presence during COVID-19 Outbreak: A Multi-Mediation Approach. *Int. J. Environ. Res. Public Health* 2020, 17, 2954. [CrossRef] [PubMed]